Author's response to reviews

Title: Dietary calcium intake and mortality risk from cardiovascular disease and all causes: a meta-analysis of prospective cohort studies

Authors:

Xia Wang (wangxiaes@sdu.edu.cn)
Ying ying Ouyang (xiongying0217@gmail.com)
Jun liu (liujunjui@sina.com)
Gang Zhao (zg000507@163.com)
Wei Bao (wei.bao@nih.gov)
Hong xia Chen (hustchenhx@163.com)
Mao sheng Yan (254251509@qq.com)

Version: 4 Date: 20 August 2014

Author's response to reviews: see over
Response to editors:
Thank you for your careful review of our manuscript.
Considering that 800 mg/day corresponds to the recommended daily intake for Swedish women aged more than 50 years [38] and it helps the reader to quickly see that both low and high calcium intake is associated with increased risk of CVD, we used 800 mg/d as the reference intake. We changed the figure from using 500 mg/d to 800 mg/d. The text were be revised in accordance with this revision.